Author's response to reviews

Title: Efficacy of Topical Cobalt Chelate CTC-96 Against Adenovirus Keratoconjunctivitis in a Rabbit Model.

Authors:
- Seth P Epstein (seth.epstein@mssm.edu)
- Yevgenia Y Pashinsky (yevgenia.pashinsky@mssm.edu)
- David Gershon (dgershon@redoxpharm.com)
- Irene Winicov (iwinicov@redoxpharm.com)
- Charlie Srivilasa (charliesrivilasa7@yahoo.com)
- Katarina J Kristic (KitKat570@hotmail.com)
- Penny A Asbell (penny.asbell@mssm.edu)

Version: 4 Date: 22 May 2006

Author's response to reviews: see over
May 22, 2006  
Seth P. Epstein, M.D. 
Department of Ophthalmology  
Annenberg 22-18, Box 1183  
1 Gustave L. Levy Place  
New York, New York   10029-6574  
Tel:   (212) 241-5606  
Fax:   (212) 289-5945  
Email:  seth.epstein@mssm.edu

Elizabeth C. Moylan PhD  
Senior Assistant Editor  
BioMed Central  
Middlesex House  
32-42 Cleveland Street  
London  
W1T 4LB  
UK

Tel: +44 (0)20 7631 9946  
Editorial Office:  http://www.biomedcentral.com/bmcophthalmol  
Email:  editorial@biomedcentral.com

Dear Dr. Moylan,

Attached, please find our edited manuscripts:  Manuscript 2126963171799982 “In Vitro Efficacy of Cobalt Chelate CTC-96 Against Adenovirus.” and Manuscript 5411825379999725 “In Vitro Efficacy of Cobalt Chelate CTC-96 Against Adenovirus Keratoconjunctivitis in a Rabbit Model.”, representing individual original research done in our laboratory. As previously mentioned, in response to the reviewers comments, the above mentioned two documents have been edited and combined into one manuscript whose title has been slightly edited to reflect the addition of the in vitro material. We propose the title of the new document be: “Efficacy of Topical Cobalt Chelate CTC-96 Against Adenovirus in a Cell Culture Model and Against Adenovirus Keratoconjunctivitis in a Rabbit Model.”

We have made all the revisions the editorial production team recommended.

We thank you for your careful consideration of this paper for publication in BioMed Central Ophthalmology. Please contact us if we can be of help in the review process.

Sincerely,
Seth P. Epstein, M.D.
Assistant Professor of Ophthalmology
Telephone:   (212) 241-5606
Fax:         (212) 289-5945
Email:       seth.epstein@mssm.edu
Corresponding author: Seth Epstein, MD,
Mount Sinai School of Medicine, Box 1183,
1 Gustave L. Levy Place,
New York, New York 10029-6574.
Telephone: (212) 241-5606.
Fax: (212) 289-5945.
Email: seth.epstein@mssm.edu